Author's response to reviews

Title: Reporting of euthanasia and physician-assisted suicide in the Netherlands: descriptive study

Authors:

Hilde Buiting (h.buiting@erasusmc.nl)
Johannes van Delden (J.J.M.vanDelden@umcutrecht.nl)
Bregje Onwuteaka-Philipsen (b.philipsen@vumc.nl)
Judith Rietjens (j.rietjens@erasusmc.nl)
Mette Rurup (m.rurup@vumc.nl)
Donald van Tol (D.G.van.Tol@med.umcg.nl)
Joseph Gevers (j.k.gevers@amc.uva.nl)
Paul van der Maas (p.vandermaas@erasusmc.nl)
Agnes van der Heide (a.vanderheide@erasusmc.nl)

Version: 7 Date: 20 October 2009

Author's response to reviews: see over
Dear editor,

Thank you very much that BMC Medical Ethics will publish our manuscript; “Reporting of euthanasia and physician-assisted suicide in the Netherlands: descriptive study”.

I have made some small changes for the Tables.
- The titles and footnotes are displaced just below the Tables itself.